Title: Transient detectable viremia and the risk of viral rebound in patients from the Swiss HIV Cohort Stud

Authors:

Jim Young (james.young@usb.ch)
Martin Rickenbach (Martin.Rickenbach@chuv.ch)
Alexandra Calmy (Alexandra.Calmy@hcuge.ch)
Enos Bernasconi (Enos.Bernasconi@eoc.ch)
Cornelia Staehelin (Cornelia.Staehelin@insel.ch)
Patrick Schmid (Patrick.Schmid@kssg.ch)
Matthias Cavassini (Matthias.Cavassini@chuv.ch)
Manuel Battegay (Manuel.Battegay@usb.ch)
Guldrych Günthard (huldrych.guenthard@usz.ch)
Heiner Bucher (heiner.bucher@usb.ch)

Version: 2
Date: 4 March 2015

Author's response to reviews:

Re-submitted with:

(1) Conclusion section added.
(2) Ethics approval statement added to the first paragraph of the Methods and an electronic citation added with links to signed documents.
(3) Email address for all authors given on title page.
(4) Figures removed from main manuscript. See separate TIFF files. The figure legends remain in the main manuscript because I assume these must be printed in the journal's font.